Author's response to reviews


Authors:

Sigrid Bjornelv (sigrid.bjornelv@ntnu.no)
Stian Lydersen (stian.lydersen@ntnu.no)
Arnstein Mykletun (arnstein.mykletun@psyhp.uib.no)
Turid Lingaas Holmen (turid.lingaas.holmen@ntnu.no)

Version: 18 Date: 25 August 2007

Author's response to reviews: see over
Dear Editor,

Thank you for accepting the manuscript “Changes in BMI-distribution from 1966-69 to 1995-97 in adolescents. The Young-HUNT study, Norway.” for publication in BMC Public Health. This manuscript was accepted for publishing in BMC Public Health, pending formatting requirements. This manuscript is revised according to the formatting changes requested:

- The authors’ names are identical in the manuscript and in the submission system.
- The headings and subheadings are written in sentence text.
- The additional files are removed, and the tables are pasted into the end of the document text file.
- Reference List is changed to References.
- The figure and table placement text are removed.
- I have also corrected the text twice. On page 3 (Background) the text was “surveying Medline 1994-I 2007,” is corrected to “surveying Medline 1994 – May 2007. Under the discussion (p10 in my version) the letter n was left alone with no meaning, and is removed.

Yours sincerely

Sigrid Bjørnelv